Author's response to reviews

Title: Radiofrequency Ablation of Liver Lesions: Quantitative Assessment of Treatment Completeness through CT Image Processing.

Authors:

Katia Passera (katiamarina.passera@marionegri.it)
Sabrina Selvaggi (sabrina.selvaggi@mail.polimi.it)
Davide Scaramuzza (davide.scaramuzza@istitutotumori.mi.it)
Francesco Garbagnati (francesco.garbagnati@istitutotumori.mi.it)
Daniele Vergnaghi (daniele.vergnaghi@istitutotumori.mi.it)
Luca Mainardi (luca.mainardi@biomed.polimi.it)

Version: 2 Date: 14 June 2012

Author's response to reviews:

Dear Editor,

we are submitting a revised of the manuscript "Radiofrequency Ablation of Liver Lesions: Quantitative Assessment of Treatment Completeness through CT Image Processing" by K. Passera et al. for consideration of your Journal. The manuscript was reviewed according to the Referees' comments and we ope to have clarify all doubts they pointed out.

We realized that Reviewers positively judged the novelty of the manuscript and recognize its significance. However, they evidenced some unclear points in our work. We hope this new version will clarify all the doubtful points.

Kindly regards

The Authors